Author's response to reviews

Title: Children's Acceptance of Milk with Xylitol or Sorbitol for Dental Caries Prevention

Authors:

Jorge L Castillo (jcastillo_1@terra.com.pe)
Peter Milgrom (dfrc@u.washington.edu)
Susan E Coldwell (scoldwel@u.washington.edu)
Ramon Castillo (rcmcasti@terra.com.pe)
Rocio Lazo (rciolazo@yahoo.com)

Version: 3 Date: 14 July 2005

Author's response to reviews:

All changes made as requested in your email:

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names.

Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers. For example:
Enitan Sawyerr1,2, Emma Veitch3
1 Plant Physiology, Zentrum fur Molekularbiologie der Pflanzen (ZMBP), Universitat Tubingen, Auf der Morgenstelle1, D-72076 Tubingen, Germany
2 Current address: Protein Interaction laboratory, Dept. Biochemistry and Molecular Biology, University of Southern Denmark, Campusvej 55, 5230 Odense, Denmark
3 Plant Biology, University of Minnesota, 220 Biological Sciences Center, 1445 Gortner Ave., St. Paul, MN55108-1095, USA

Keywords section - Please remove the keywords section from the manuscript
Page breaks - Please remove page breaks from the manuscript (excluding the title page)
Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).